Author's response to reviews

Title: Screening and prophylactic treatment of chronic renal failure patients colonized with Staphylococcus aureus: a systematic review and meta-analysis

Authors:

Cibele Grothe PhD (cibelegrothe@hotmail.com)
Mónica Taminato PhD (mo_tami@yahoo.com.br)
Angélica Belasco PhD (abelasco@unifesp.br)
Dulce Barbosa PhD (dulce.barbosa@unifesp.br)
Ricardo Sesso PhD (Rsesso@unifesp.br)

Version: 2
Date: 6 August 2014

Author's response to reviews:

Revisions in this version:

1. Line numbering - Manuscript revised to include line numbers.
2. Authors Contributions - Authors' contributions added to the cover letter as requested.
3. Figures - Figures removed the main body of the manuscript and upload as figure files. I also removed the headings from the images and provided figure legends in the manuscript as detailed.